CS3233 Homework 2

Due on 9 February 2002 at 9:55am. 100 marks Topics: DP Problems on the Spanish Judge

1. Written Homework on DP For a sample expectation, please refer to the question and sample solution on the ”Litte Shop of Flowers”.

Each of you are assigned the following questions to do:

ariyarat 104, 166, 323 lowchinh 231, 116, 147
chewbakt 108, 111, 430 stevenha 348, 357, 435
daibingt 164, 166, 323 wuyongzh 323, 430, 435
eng11963 437, 442, 526 zhangkai 562, 580, 590
fengchun 607, 624, 662 zhengzhi 672, 674, 702
jimmobar 707, 709, 714 johnadri 103, 231, 108

Each solution is worth 10 marks. [30]

2. Programming Assignment on DP Kindly complete any 7 of the above questions on the Spanish Judge. Each solution is worth 10 marks. [70]

[Self Practice ] Other DP questions on the Spanish Judge 10003, 10048, 10051, 10066, 10069, 10074, 10081, 10086, 10091, 10100, 10130, 10131, 10133, 10154, 10163.